16 November, 2016

World Journal of Gastroenterology

Dear Editor,

Thank you for the opportunity to improve and resubmit our article. We have addressed all issues raised by you and the reviewers and

made all necessary amendments.

I confirm that all authors have seen and agree with the contents of

the revised manuscript.

We look forward to your assessment.

Sincerely,

Giorgos Chouliaras